

## **Comments of the reviewer and the point to point answers**

**Point 1:**The hypothesis / concepts being proposed are reasonable but I was hoping for more experimental data e.g. in rat studies to provide some stronger support. Is there further published / unpublished data available that can be described? There is mention of preliminary experiments and there are some comments about experimental studies, but the experimental data seems very limited.

**Response:**This comment is very important to me. This paper is the concept of our opinion.we will describe more experimental data in further.

**Point 2:** There is repetition of a paragraph between the Introduction and Hypothesis sections.

**Response:**I have rewritten it and delete the repetition of the paragraph.